Author's response to reviews

Title: Smoking, environmental tobacco smoke and risk of renal cell cancer: a population-based case-control study

Authors:

Ryan P. Theis (rtheis@ichp.ufl.edu)
Suzanne M. Dolwick Grieb (sdolwick@ufl.edu)
Deborah Burr (burr@stat.ufl.edu)
Tariq Siddiqui (siddit@medicine.ufl.edu)
Nabih R. Asal (asal@phhp.ufl.edu)

Version: 6 Date: 20 December 2008

Author's response to reviews:

Dear Editors,

Attached please find our third revision of the manuscript, "Smoking, Environmental Tobacco Smoke, and Renal Cell Cancer: A Population-Based Case-Control Study".

This revision includes minor typographical changes:
- pg. 6, 1st full paragraph: hyphen removed from "in-person:
- pg. 8, revised to "an effect that was likely due to misclassification..."
- pg. 13, 1st full paragraph, 1st sentence now reads: "While this study found increased RCC risks among ever-smokers who reported inhaling, these associations were attenuated after controlling for pack-years." This change was made to reflect language used in the results for inhalation patterns.
- Table 1. Percentages were set to one decimal place.

Sincerely,

Ryan Theis